Author's response to reviews

Title: Culture, Acculturation and Tobacco Use in Hmong, Khmer, Laotians, and Vietnamese Communities in Minnesota

Authors:

Diana J Burgess (Diana.Burgess@va.gov)
Jeremiah Mock (jeremiah.mock@gmail.com)
Barbara A. Schillo (BSchillo@clearwaymn.org)
Jessie Saul (JSaul@naquitline.org)
Tam Phan (dophan@chorus.net)
Yanat Chhith (yanat.chhith@aapress.com)
Nina Alesci (Nina_Alesci@bluecrossmn.com)
Steven. Foldes (steven.foldes@gmail.com)

Version: 6
Date: 18 July 2014

Author's response to reviews: see over
July 18, 2014

Dr. Natalie Pafitis
Executive Editor, BMC Public Health

RE: MS 4330763061072564

Dear Dr. Pafitis:

On behalf of my co-authors, I am resubmitting a revised manuscript for publication in *BMC Public Health* entitled, “Culture, Acculturation and Smoking in Hmong, Khmer, Laotians, and Vietnamese Communities in Minnesota.” This version of the manuscript includes an ethics approval statement in the Methods section (p. 5).

“This study was approved by the Institutional Review Board of the Minnesota Department of Health.”

Please address all correspondence concerning this manuscript to Diana Burgess, Ph.D. (One Veterans Drive; Minneapolis, MN 55419; Phone: 612-467-1591; Fax: 612-467-5699; Diana.Burgess@va.gov).

Thank you for the opportunity to resubmit our revised manuscript to *BMC Public Health*.

Sincerely,

Diana Burgess, Ph.D.
Associate Professor, Department of Medicine